# Improving care transfers for homeless patients after hospital discharge: a realist evaluation

Michelle Cornes,<sup>1\*</sup> Robert W Aldridge,<sup>2</sup> Elizabeth Biswell,<sup>1</sup> Richard Byng,<sup>3</sup> Michael Clark,<sup>4</sup> Graham Foster,<sup>5</sup> James Fuller,<sup>1</sup> Andrew Hayward,<sup>2</sup> Nigel Hewett,<sup>6</sup> Alan Kilmister,<sup>1</sup> Jill Manthorpe,<sup>1</sup> Joanne Neale,<sup>7</sup> Michela Tinelli<sup>4</sup> and Martin Whiteford<sup>8</sup>

<sup>1</sup>Health and Social Care Workforce Research Unit, King's College London, London, UK <sup>2</sup>Institute of Health Informatics, University College London, London, UK

**Declared competing interests of authors:** Michael Clark was a member of the COVID-19 reviewing committee (2020–1) and Prophylaxis Platform Study Funding Committee (25 November 2020). Graham Foster reports consultancy and speaker fees from Gilead Sciences (Foster City, CA, USA), F. Hoffman-La Roche Ltd (Basel, Switzerland), uniQure (Amsterdam, the Netherlands) and BioMarin Pharmaceutical (Novato, CA, USA). Andrew Hayward reports membership of the National Institute for Health Research Health Technology Assessment Antimicrobial Resistance Themed Call Board (2013–14).

Published September 2021 DOI: 10.3310/hsdr09170

<sup>&</sup>lt;sup>3</sup>Clinical Trials and Health Research, University of Plymouth, Plymouth, UK

<sup>&</sup>lt;sup>4</sup>Care Policy and Evaluation Centre, London School of Economics and Political Science, London, UK

<sup>&</sup>lt;sup>5</sup>Blizard Institute, Queen Mary University of London, London, UK

<sup>&</sup>lt;sup>6</sup>Pathway and the Faculty for Homeless and Inclusion Health, London, UK

<sup>&</sup>lt;sup>7</sup>National Addiction Centre, Institute of Psychiatry, Psychology & Neuroscience, King's College London, London, UK

<sup>&</sup>lt;sup>8</sup>Department of Community Nursing and Community Health, Glasgow Caledonian University, Glasgow, UK

<sup>\*</sup>Corresponding author michelle.cornes@kcl.ac.uk In memory of Darren O'Shea (1977–2021).

## **Plain English summary**

Care transfers for homeless patients after hospital discharge Health Services and Delivery Research 2021; Vol. 9: No. 17

DOI: 10.3310/hsdr09170

NIHR Journals Library www.journalslibrary.nihr.ac.uk

## **Plain English summary**

n 2013, it was reported that 70% of patients who were homeless were discharged from hospital to the streets without having their care and support needs addressed. To stop this from happening, the government passed £10M to community and voluntary organisations to develop specialist homeless hospital discharge schemes. This money could also be spent on 'step-down' intermediate care (i.e. places where people can stay for a short time while they recover).

This report describes the evaluation of these schemes. The evaluation lasted from 2015 to 2019. The research involved listening to the stories and experiences of 70 homeless patients who were followed up for 3 months after their discharge from hospital. We linked information about 3882 patients to a national collection of information at NHS Digital. This allowed us to find out whether homeless patients had been to accident and emergency departments or had gone back into hospital after they had been discharged. We also looked at how much these specialist services cost and if they were a good use of public money. Our research team included people with experience of being homeless themselves.

We found that these schemes were very effective in supporting people to leave hospital safely, especially if they provided practical help and stayed in touch with people until other support arrived. Schemes that employed specialist general practitioners and nurses who had a good understanding of homelessness also helped people get the health care they needed after hospital. This was very important to keep people healthy and prevent them dying unnecessarily. We also discovered that homeless patients who received step-down support made fewer visits to accident and emergency (18% less) than those who did not. Once the pilot funding ended, some of the schemes struggled to continue and the positive benefits evaporated.

### **Health Services and Delivery Research**

ISSN 2050-4349 (Print)

ISSN 2050-4357 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HS&DR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Health Services and Delivery Research journal

Reports are published in *Health Services and Delivery Research* (HS&DR) if (1) they have resulted from work for the HS&DR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **HS&DR** programme

The HS&DR programme funds research to produce evidence to impact on the quality, accessibility and organisation of health and social care services. This includes evaluations of how the NHS and social care might improve delivery of services.

For more information about the HS&DR programme please visit the website at https://www.nihr.ac.uk/explore-nihr/funding-programmes/health-services-and-delivery-research.htm

#### This report

The research reported in this issue of the journal was funded by the HS&DR programme or one of its preceding programmes as project number 13/156/10. The contractual start date was in October 2015. The final report began editorial review in February 2020 and was accepted for publication in May 2021. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HS&DR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care.

Copyright © 2021 Cornes et al. This work was produced by Cornes et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

#### NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

#### **NIHR Journals Library Editors**

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk